

RESEARCH ARTICLE

10.1029/2023MS004007

Manuel Saigger and Tobias Sauter are first authors.

Key Points:

- We introduce a new module to simulate drifting snow within the framework of the Weather Research and Forecasting (WRF) model
- The module is tested in an idealized setting for physical consistency
- We show that local sublimational cooling has larger-scale non-local effects on the wind field

Correspondence to:

M. Saigger and T. Sauter, manuel.saigger@fau.de; tobias.sauter@geo.hu-berlin.de

Citation:

Saigger, M., Sauter, T., Schmid, C., Collier, E., Goger, B., Kaser, G., et al. (2024). A drifting and blowing snow scheme in the weather research and forecasting model. *Journal of Advances in Modeling Earth Systems*, *16*, e2023MS004007. https://doi.org/10.1029/ 2023MS004007

Received 11 SEP 2023 Accepted 5 MAY 2024

Author Contributions:

Formal analysis: Manuel Saigger, Christina Schmid Funding acquisition: Tobias Sauter, Georg Kaser, Thomas Mölg Software: Manuel Saigger, Tobias Sauter, Christina Schmid, Emily Collier Writing – original draft: Manuel Saigger, Tobias Sauter, Thomas Mölg Writing – review & editing: Manuel Saigger, Tobias Sauter, Christina Schmid, Emily Collier, Brigitta Goger, Georg Kaser, Rainer Prinz, Annelies Voordendag, Thomas Mölg

© 2024 The Authors. Journal of Advances in Modeling Earth Systems published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

A Drifting and Blowing Snow Scheme in the Weather Research and Forecasting Model

Manuel Saigger¹, Tobias Sauter², Christina Schmid¹, Emily Collier³, Brigitta Goger^{3,4}, Georg Kaser³, Rainer Prinz³, Annelies Voordendag^{3,5}, and Thomas Mölg¹,

¹Climate System Research Group, Institute of Geography, Friedrich-Alexander-University Erlangen-Nürnberg (FAU), Erlangen, Germany, ²Insitute of Geography, Humboldt-Universität zu Berlin, Berlin, Germany, ³Department of Atmospheric and Cryospheric Sciences (ACINN), University of Innsbruck, Innsbruck, Austria, ⁴Center for Climate Systems Modeling, ETH Zürich, Zürich, Switzerland, ⁵Institute of Geodesy and Photogrammetry, ETH Zürich, Zürich, Switzerland

Abstract Wind-driven redistribution of surface snow is one of the key factors leading to heterogeneous accumulation of snow at small scales. Understanding the processes that lead to this heterogeneous accumulation is, therefore, of great importance to many glaciological and hydrological questions. High-quality information on the wind field is necessary to realistically represent drifting snow. Here, we introduce a novel, intermediate-complexity drifting and blowing snow module for the Weather Research and Forecasting (WRF) model that integrates seamlessly into the standard WRF infrastructure. The module also accounts for snow particle sublimation and considers the thermodynamic feedback on the atmospheric fields. In an idealized model environment the module was tested for physical consistency. Sensitivity experiments showed that drifting snow sublimation has on the one hand local effects on the erosion and deposition patterns and on the other hand non-local effects on the larger-scale atmosphere. The simple and computationally efficient implementation allows this module to be used for small-scale and large-scale simulations in polar and glaciated regions.

Plain Language Summary Transport of snow by the wind can have high impact on local glacier mass changes as it leads to non-uniform amounts of snow on the ground. In order to simulate and better understand this process we introduce a new modeling framework that is included into a widely used atmospheric model. Test simulations and sensitivity experiments show the physical consistency of the model. Complex interactions between different processes like snow erosion, drifting snow sublimation and the wind field show the necessity of coupling the snow and atmospheric models.

1. Introduction

The spatial and temporal variability of snow cover in complex mountain areas is an important component of the water budget of hydrological catchments and one of the key drivers for the mass balance of alpine glaciers. Larger-scale orographic enhancement of precipitation (Houze, 2012) as well as small-scale micro-physical processes and interactions between the near-surface flow field and hydrometeors (often referred to as preferential deposition) lead to heterogeneities in the initial accumulation of precipitation (e.g., Gerber et al., 2019; Lehning et al., 2008; Mott et al., 2014; Vionnet et al., 2017; Zängl, 2007, 2008). Redistribution of snow by the wind during and after the precipitation event further alters the small-scale variability in snow accumulation. Mott et al. (2011) demonstrated that variability in the snow accumulation at the ridge scale, in the order of tens of meters, is dominated by drifting snow processes and preferential deposition. Topographic features such as ridges and gaps shape the small-scale heterogeneity by locally accelerating the flow so that shear stresses and turbulence kinetic energy become strong enough to effectively remove snow from the surface (Mott et al., 2018). Turbulent motions transport the eroded snow away from the surface so that the particles can be advected over long distances with the background flow. During transport the particles are subject to gravity-driven subsidence and will accumulate in wind-protected areas. Depending on the atmospheric conditions, drifting snow particles often lose mass by sublimation, which in consequence modifies the vertical structure of temperature and moisture.

Several studies showed that redistribution of snow is essential to understand the spatial structure of local snow accumulation on timescales ranging from single events to entire seasons (Bernhardt et al., 2012; Musselman et al., 2015; J. W. Pomeroy & Gray, 1995; J. W. Pomeroy et al., 1998; Schirmer et al., 2011; Vionnet et al., 2021). Different studies implied that their calculations of glacier mass balance can be improved by accounting for drifting snow processes in the simulations (Abrahim et al., 2023; Mortezapour et al., 2020;



Pradhananga & Pomeroy, 2022; Temme et al., 2023; Terleth et al., 2023). The importance of drifting snow sublimation was found to be highly variable depending on the environment and the flow conditions (e.g., Groot Zwaaftink et al., 2011; Groot Zwaaftink et al., 2013; Le Toumelin et al., 2021; Lenaerts & Van den Broeke, 2012; Lenaerts et al., 2010; MacDonald et al., 2010; Sauter et al., 2013; Strasser et al., 2008). Apart from alpine contexts, polar ice bodies have been shown to be greatly influenced by drifting snow on the one hand due to the redistribution and drifting snow sublimation (e.g., Amory et al., 2021; Lenaerts et al., 2010), but also by alternating the radiative budget (Le Toumelin et al., 2021). Nevertheless, both the measurement and modeling of drifting snow remain challenging due to the very small-scale nature and the complex interplay of many simultaneously acting processes.

Snow drift schemes and parameterizations range from simple heuristic solutions to complex models. Computationally expensive numerical approaches are designed to resolve the relevant physical processes either in a column-integrated form (e.g., Déry & Yau, 1999; Liston & Sturm, 1998; J. Pomeroy et al., 1993) or fully threedimensional (e.g., Marsh, Pomeroy, Spiteri, & Wheater, 2020; Marsh, Pomeroy, & Wheater, 2020; Sharma et al., 2023; Vionnet et al., 2014). Other authors developed statistical parameterization schemes based on interpolated wind fields adjusted for topographic features (e.g., Durand et al., 2005; Winstral & Marks, 2002). In terms of the representation of the snow pack a wide range of complexity has been employed as well. Snow drift schemes have been coupled to highly complex grain-scale models like SNOWPACK (Lehning et al., 2008; Sharma et al., 2023) or CROCUS (Vionnet et al., 2014), to multi-layer models like SISVAT (Amory et al., 2021), as well as to simpler models such as the Community Land Model (CLM, Xie et al., 2019) or the bulk-model SnowTran-3D (Liston et al., 2007).

In recent years, there have been extensive modeling efforts with strong emphasis on flat and wide open areas, such as the arctic tundra (Déry & Yau, 2001; Liston & Sturm, 1998; J. Pomeroy & Gray, 1990), or large ice masses (Amory et al., 2021; Bintanja, 2000; Jaedicke, 2002; Lenaerts & Van den Broeke, 2012). Several studies investigated the effect of snow drift in alpine environments, however, due to the highly complex winds most of these studies relied on either terrain-adjusted interpolated wind fields (e.g., Schirmer et al., 2011; Strasser et al., 2008; Winstral & Marks, 2002; Winstral et al., 2013) or on wind-library approaches either with precomputed semi-idealized wind profiles (e.g., Bernhardt et al., 2012; Dadic et al., 2010; Groot Zwaaftink et al., 2011; Mott & Lehning, 2010; Mott et al., 2010) or downscaling tools like WindNinja or WindMapper (Forthofer et al., 2014; Marsh et al., 2023; Vionnet et al., 2021). Very recently, Haddjeri et al. (2023) successfully applied the deep-learning based downscaling tool DEVINE to drifting snow simulations. In the last years, nested high-resolution simulations in real-world settings have provided valuable insights into, for example, the dynamics of down-slope windstorms (Umek et al., 2021), precipitation mechanisms in highly complex terrain (Gerber et al., 2018; Vionnet et al., 2017) or surface exchange processes over glaciated areas (Goger et al., 2022). The inclusion of high-quality wind fields provided by such approaches can improve our understanding also for drifting snow in alpine environments. Sharma et al. (2023) and Vionnet et al. (2014) provided high-complexity modeling frameworks for such investigations.

In our study, however, we want to introduce a new, intermediate-complexity framework to simulate drifting snow. Our approach makes use of the already existing capabilities of the Weather Research and Forecasting (WRF) model including a standard land-surface model (Noah-MP) for the treatment of snow. Apart from that, the tracer module is employed for particle transport, hence all WRF-internal schemes and options for advection and turbulent diffusion can be used. With this intermediate-complexity design we are able to make use of high-quality wind fields and a reasonable representation of the snow pack while keeping the computational costs rather low, which will support our future goals of investigations on climatic timescales. This paper describes the physical and numerical fundamentals of the scheme and analyses the skill and physical consistency of this intermediate-complexity snow drift scheme. In our paper, we will use the term drifting snow for all wind-driven transport of snow without distinguishing additionally between drifting and blowing snow.

The remainder of this paper will be structured as follows: Section 2 gives an overview on the snow drift scheme and its implementation in WRF. Section 3 introduces the setup for the idealized test environment, of which the results are presented in Section 4. Finally, Section 4.5 gives an outlook for future real-world applications where we apply the module to a drifting snow event in the Alps and where we can verify the modeled results by comparing them to snow depth changes observed by Terrestrial Laser Scanning.



2. Model Equations and Implementation

The snow drift scheme was implemented in the Weather Research and Forecasting (WRF) model version 4.1.2 (https://github.com/wrf-model/WRF/releases/tag/v4.1.2 Skamarock et al., 2019). The latest snow drift scheme code is available at Saigger (2024).

2.1. Snow Drift Scheme

The implemented snow drift scheme is based on the work of Sauter et al. (2013).

We employ the widely used approach of dividing the process of drifting snow into a saltation layer and snow particles in suspension (Mott et al., 2018). The saltation layer is fully parameterized and acts as a lower boundary condition for the flux of snow into the atmosphere, while snow particles in suspension are transported by the mean wind and turbulent diffusion and are subject to gravity-driven subsidence as well as to sublimation. This scheme therefore treats drifting of snow as continuous phases that interact only with the background flow while neglecting particle interactions (Bintanja, 2000; Durand et al., 2005; Gauer, 2001; Liston & Sturm, 1998; Naaim et al., 1998; J. Pomeroy & Gray, 1990; Sauter et al., 2013; Schneiderbauer et al., 2008). According to the continuum equation for mass conservation, the local rate of snow concentration change $\partial \phi_s / \partial t$ [kg m⁻³ s⁻¹] is given by

$$\underbrace{\frac{\partial \phi_s}{\partial t}}_{(I)} + \underbrace{\frac{\partial (\phi_s u_i)}{\partial x_i}}_{(II)} = \underbrace{-\frac{\partial}{\partial x_i} \left(\overline{\phi_{r_s} u_{r_i}}\right)}_{(III)} + \underbrace{\frac{\partial}{\partial x_3} (V\phi_s)}_{(IV)} + \underbrace{\left(\frac{\partial \phi_s}{\partial t}\right)_{sub}}_{(V)}.$$
(1)

Here, (II) describes the advection by the mean wind with x_i the cartesian coordinates and u_i the cartesian components of the wind vector. The term (III) describes the turbulent flux divergence, (IV) the downward flux of snow particles due to gravity and (V) the concentration loss due to drifting snow sublimation. Advection and turbulent diffusion are handled by WRF-internal modules for the transport of passive tracers, while we parameterize terms (IV) and (V) as described in the following. The fallout velocity

$$V(z) = -\frac{A}{r(z)}\sqrt{\left(\frac{A}{r(z)}\right)^2 + B \cdot r(z)}$$
(2)

at height z [m] is following Vionnet et al. (2014) with the two parameters

$$A = \frac{6.203 \cdot \nu_{air}}{2} \tag{3}$$

and

$$B = \frac{5.516 \cdot \rho_{ice}}{4 \cdot \rho_{air}} \cdot g. \tag{4}$$

Here ν_{air} [m² s⁻¹] represents the viscosity of air, ρ_{ice} and ρ_{air} [kg m⁻³] the density of ice and air, respectively, and g [m s⁻²] the acceleration due to gravity. The height-dependent mean particle radius r(z) [m] is calculated analogous to Sauter et al. (2013) with

$$r(z) = r_0 \cdot z^{-0.258},\tag{5}$$

where r_0 [m] is the particle radius at ground level following Gordon et al. (2010) with:

$$r_0 = 0.5 \left(\frac{7.8 \cdot 10^{-6} u_{\star}}{0.036} + 31 \cdot 10^{-6} \right),\tag{6}$$

and u_{\star} [m s⁻¹] the friction velocity. Optionally, V(z) and r_0 can be set to constant values in the namelist settings.



9422466, 2024, 6, Downle

from https:

//agupubs.onlinelibrary.wiley.com/doi/10.1029/2023MS004007 by ETH Zurich, Wiley Online Library on [12/06/2024]. See the Terms

and C

Term (V) of Equation 1 accounts for the mass loss of suspended snow by sublimation and is calculated based on the formulation of Thorpe and Mason (1966) that has widely been used in earlier studies (e.g., Bintanja, 2000; Déry & Yau, 1999; Gallée et al., 2001; Sauter et al., 2013; Sharma et al., 2023; Vionnet et al., 2014). Sublimation from particles in the saltation layer is neglected similar to earlier modeling approaches. The sublimation-loss rate of suspended snow is approximated by $\psi_s \phi_s$ [kg m⁻³ s⁻¹], where ψ_s [s⁻¹] is the sublimation-loss rate coefficient describing the change of snow particle mass due to heat exchange and ventilation effects while neglecting radiation effects:

$$\psi_s = \frac{3}{4\pi\rho_{ice}r^3} \frac{2\pi r\sigma}{\frac{L_s}{KNaT_{air}} \left(\frac{L_s}{R_s T_{air}} - 1\right) + R_v \frac{T_{air}}{ShDe_i}}.$$
(7)

Here σ is the water vapor deficit with respect to ice, L_s the latent heat of sublimation (2.838 \cdot 10⁶ J kg⁻¹), K the thermal conductivity of air (0.024 W m⁻¹ K⁻¹), R_v the gas constant for water vapor (461.5 J kg⁻¹ K⁻¹), D the molecular diffusivity of water vapor in air (2.25 \cdot 10⁻⁵ m² s⁻¹), T_{air} the air temperature, and e_i the saturation vapor pressure with respect to ice. The Nusselt number and Sherwood number Nu and Sh describe the transfer of heat and mass at the particle surface with

$$Nu = Sh = \begin{cases} 1.79 + 0.0606\sqrt{Re}, \text{Re} \le 10\\ 1.88 + 0.580\sqrt{Re}, \text{Re} > 10 \end{cases}$$
(8)

where $\operatorname{Re}(z)$ is the particle Reynolds number:

$$\operatorname{Re}(z) = \frac{2rV}{\nu_{air}}.$$
(9)

The model therefore reflects the ventilation effect of the falling snow particles, ventilation due to turbulence is neglected. The scheme considers the effect of sublimation on the vertical profiles of temperature and humidity. For that, the local air temperature is reduced by the heat required for sublimation and the local specific humidity is increased accordingly. This feedback self-limits the sublimation process since its intensity depends on the saturation deficit and the air temperature of the environment (Sauter et al., 2013).

As described above, the drifting snow is divided into a saltation layer and the suspended snow particles. The snow mass concentration within the saltation layer is primarily gained by the aerodynamic entrainment of snow particles from the underlying snowpack. Once the surface shear stress exceeds the inertia and the cohesive bonds of the snow particles, transport takes place. Therefore, the vertical erosional mass flux from the surface into the saltation layer q_e [kg m⁻² s⁻¹] is assumed to be proportional to the excess surface shear stress (Schmidt, 1980) with

$$q_e = e_{salt}\rho_{air} \left(u_{\star}^2 - u_{th}^2\right). \tag{10}$$

Here, ρ_{air} [kg m⁻³] is the air density and u_{th} [m s⁻¹] the friction threshold velocity. The heuristic parameter e_{salt} [–] controls the efficiency of the erosion process. The threshold friction velocity u_{th} is proportional to the snow density ρ_{snow} [kg m⁻³] following Walter et al. (2004):

$$u_{th} = 0.0195 + \left(0.021\sqrt{\rho_{snow}}\right). \tag{11}$$

Similar to Sauter et al. (2013) the drag exerted by the particles on the flow field is accounted for by limiting the saltation-layer snow concentration ϕ_{salt} to a maximum value $\phi_{salt,max}$ (J. Pomeroy & Gray, 1990):

$$\phi_{salt,max} = \frac{\rho_{air}}{3.29 \cdot u_*} \left(1 - \frac{u_{th}^2}{u_*^2} \right).$$
(12)

on Wiley Online

use; OA article

are governed by the applicable Creat



Additionally, a drag-corrected friction velocity $u_{\star c}$ following Naaim et al. (1998) is introduced that reduces the effective friction velocity to u_{th} in the case of a fully saturated saltation layer:

$$u_{\star,c} = u_{\star} + (u_{th} - u_{\star}) \left(\frac{\phi_{salt}}{\phi_{salt,max}}\right)^2.$$
(13)

Replacing u_{\star} by $u_{\star,c}$ in Equation 10 results in the drag-corrected erosion flux $q_{e,c}$

$$q_{e,c} = e_{salt}\rho_{air} \left(\left(u_{\star} + (u_{th} - u_{\star}) \left(\frac{\phi_{salt}}{\phi_{salt,max}} \right)^2 \right)^2 - u_{th}^2 \right).$$
(14)

Following J. Pomeroy and Gray (1990) the height of the saltation layer h_{salt} is described as

$$h_{salt} = \frac{1.6 \cdot u_*^2}{2g}.$$
 (15)

With the snow mass added to the saltation layer by $q_{e,c}$ and h_{salt} the snow concentration in the saltation layer is updated. The saltation layer then serves as a reservoir for the flux of snow into the atmosphere f_{salt} [kg m⁻² s⁻¹]. Similar to Gallée et al. (2001) f_{salt} is described as a bulk flux:

$$f_{salt} = UC_D(\phi_{salt} - \phi_{s,1}), \tag{16}$$

where $U \,[\mathrm{m \, s^{-1}}]$ and $\phi_{s,1} \,[\mathrm{kg \, m^{-3}}]$ are the horizontal wind speed and the snow particle concentration at the lowest model level of WRF, respectively. C_D is the bulk transfer coefficient for momentum as calculated within the surface layer scheme.

As the friction velocity drops below the threshold, deposition takes place. Similar to Beyers et al. (2004) the deposition flux q_d [kg m⁻² s⁻¹] corresponds to the downward flux at the lowest model level z_1 and the modified shear stress ratio

$$q_d = V(z_1)\phi_{s,1} \cdot \max\left(\frac{u_{th}^2 - u_*^2}{u_{th}^2}, 0\right).$$
(17)

2.2. Implementation of the Snow Drift Scheme in the WRF Model

(

The implementation of the snow drift scheme consists of three basic modules, which include a module for erosion (module_snower.F), sublimation (module_snowsubl.F) and deposition (module_snowset.F) of snow particles as well as an auxiliary module for the calculation of the particle radius and the fall velocity (module snowvel.F). The four modules are called by a high-level driver (module_snowdriver.F). The scheme is executed each time step between the land surface model (LSM) and the planetary boundary layer (PBL) scheme. The transport of snow particles is implemented as a passive tracer. This allows the user to choose between all available WRF-internal schemes and options for advection and turbulent diffusion. For real-case applications, the scheme requires the use of the LSM model Noah-MP (Niu et al., 2011). Noah-MP employs a snow model with up to three layers, that allows to represent multiple processes within the snow pack. Similar to other intermediate-complexity models, Noah-MP has been shown to substantially improve snow pack simulations both on a daily and seasonal scale compared to simpler bulk approaches (Arduini et al., 2019; Niu et al., 2011) while avoiding the relatively high computational costs of high-complexity models (Sharma et al., 2023). The snow drift scheme can be activated via the WRF namelist. The modules for erosion, sublimation and deposition can be activated separately. Basic parameters for calculating particle size, fallout velocity and the saltation coefficient can also be set (Table 1). Section 4.4 summarizes the sensitivity of simulations through setting different parameter values.



WRF Namelist Options Used in the CTRL Simulation					
&physics					
sf_surface_physics = 0	Land surface model (deactivated (0), noah_mp (4))				
$snow_opt = 1$	Activate snow drift scheme				
$er_opt = 1$	Activate snow erosion				
$subl_opt = 2$	Sublimation: no feedback (1), with feedback (2), no sublimation (0)				
$dep_opt = 1$	Activate deposition				
ustlim = -	Limit u^* to constant value (optional, req. er_opt = 1)				
rgro = -	Fixed ground particle radius (optional, default: Equation 6)				
$esalt = 5 \cdot 10^{-4}$	Saltation efficiency (default $5 \cdot 10^{-4}$)				
pavel = -	Constant settling velocity (optional, default: Equation 2)				

 Table 1

 WRF Namelist Options Used in the CTRL Simulation

Before integrating the snow particle transport equation, the scheme passes through the following steps:

- 1. Calculation of the erosion flux for each snow-covered grid cell as a function of the maximum snow particle concentration in the saltation layer. The erosion flux decreases the snow on the ground and leads to the increase of the snow particle concentration in the saltation layer. (module_snower.F)
- 2. Then the mass flow between the saltation layer and the lowest level of the suspension layer is calculated. (module_snower.F)
- 3. Now the particle radius and the fallout velocity are determined. These quantities are then passed on to the sublimation module. (module_snowvel.F)
- 4. The snow drift sublimation is now calculated in each grid cell. The energy and mass fluxes due to sublimation then modify the specific humidity and air temperature within the grid cells (if feedback has been activated in the namelist). (module_snowsubl.F)
- 5. In the last step, the deposition flux is determined, which is calculated on the basis of the particle concentration and fallout velocity. The deposition flux in the lowest model layer leads to the accumulation of snow particles and thus increases the snow on the ground. (module_snowset.F)

3. Model Setup

The WRF model has been described in numerous previous papers, as it is "the world's most-used atmospheric model" (Powers et al., 2017). In brief, it is a fully non-hydrostatic numerical model and includes a large suite of physics options (e.g., radiation, boundary layers, shallow convection) as well as the possibility to conduct idealized or realistic runs. To show the capability of the scheme, we perform idealized simulations with the WRF version 4.1.2 (Section 2). The next paragraphs describe the settings for the control run (CTRL) and the sensitivity experiments in an idealized setup. An outlook toward real-case applications is presented in Section 4.5.

The model setup is inspired by the classical studies which perform idealized atmospheric simulations for downslope windstorms (Nappo, 2013) or orographic precipitation (e.g., Jiang, 2003; Kirshbaum & Durran, 2004; Mölg et al., 2009). Accordingly, we assume a Gaussian-shaped hill with a height h of

$$h(x,y) = h_m \cdot e^{\frac{-(x^2+y^2)}{a^2}},$$
(18)

where h_m is the height of the mountain peak and *a* is the mountain half-width, for which we chose 500 m and 4 km, respectively. The domain size is 500 grid cells in the *x*-direction and 300 cells in the *y*-direction, with a grid spacing of 200 m in both horizontal directions; this eventually produces a domain 100 km (W-E) by 60 km (N-S) wide (Figure 1a). The domain is larger in W-E direction as the meteorological wind input is a zonal flow (see below) and, thus, the flow has ample space to develop in front and after the obstacle. In any case is the domain size well beyond $10 \cdot a$ in the horizontal and even larger than $20 \cdot a$ in the W-E dimension (e.g., Jiang, 2003). In the





Figure 1. Overview of the model domain and initial conditions. The model domain is depicted in (a) with black contour lines of model topography with an interval of 100 m. The locations of the vertical cross section CS is indicated as a green line, the vertical profile P as a black X. The initial wind direction is indicated as black arrows in (a). The initial profiles of potential temperature, specific humidity and zonal wind are depicted in (b-d).

vertical, 73 terrain-following levels reach up to an elevation of 10 km above the surface, and the smallest layer depth (highest resolution) of 10 m at the surface increases to a maximum of 200 m about 4 km above the surface using grid stretching (e.g., Wang & Kirshbaum, 2015). Rayleigh damping is activated for the upper 5 km of the domain to prevent gravity waves from artificial reflection.

The simulations are initialized horizontally homogeneously with profiles of potential temperature, specific humidity and the zonal and meridional wind component (Figures 1b–1d). The boundary conditions employed are periodic in y-direction and open in x-direction. The initial profiles are defined by a dry buoyancy (Brunt-Väisälä) frequency (N_d) of 0.015 s⁻¹, and a velocity U of 7.5 m s⁻¹ of the zonal wind (meridional wind is zero) which results in a non-dimensional mountain height $\hat{h} = \frac{N_d h_m}{U} = 1$ (Lin, 2007). In order to avoid artifacts at the domain boundaries a logarithmic wind profile is employed for the lowest 100 m of the initial profile. The entire initial atmospheric profile has a relative humidty of 80%. In consideration of representing an alpine environment at an altitude of about 2,000 m, the surface potential temperature is set to 290 K and the surface pressure to 800 hPa at the model initialization. With regard to the snow drift process, CTRL is initialized with a snow density of 100 kg m⁻³ over the entire domain, representing freshly fallen snow, and a snow depth of 1 m which ensures that at no point in the domain the entire snow pack gets removed.

Each model run is integrated 6 hr forward with a fixed time step of 0.9 s. After 3 hr of spin-up time a quasistationary state is reached, therefore we use only the output from hours 3 to 6 for the analysis. Due to the idealized nature of the runs and in order to attribute snow depth changes to snow drift as clearly as possible, most WRF-included physics options are deactivated. This is standard in idealized simulations, and the question of which options to keep depends on the specific processes that shall be examined by the model runs. In our case, the MM5 surface layer scheme (Fairall et al., 2003) is employed. We use the MYNN level 2.5 planetary boundary layer scheme (Nakanishi & Niino, 2006), however, we keep in mind that with our horizontal grid spacing of $\Delta x = 200$ m our simulations are within the "gray zone" of turbulence (Wyngaard, 2004). The land surface model is deactivated, which means that after initialization only erosion and deposition can alternate the snow on the ground. Other processes like melt, compaction or metamorphism therefore are not taking place in this case. In the snow drift scheme, CTRL has the sublimation of drifting particles and the associated feedback on the temperature and humidity fields activated. The saltation coefficient is set to $5 \cdot 10^{-4}$ (Sauter et al., 2013) and the particle radius at the surface results from Equation 5, while the particle fall speed is calculated with Equation 2. In order to get a deeper understanding for the behavior of the model, several sensitivity experiments were conducted as will be discussed in Sections 4.3 and 4.4.

4. Results and Discussion

4.1. Mesoscale Flow Conditions

In this first subsection, we evaluate whether the modeled mesoscale flow features agree with the known physical principles of earlier studies (overviews e.g. in Durran, 1990; Jackson et al., 2013; Lin, 2007; Nappo, 2013; Smith, 1979) and whether these flow features favor drifting snow.





Figure 2. Average flow field for model time 3–6 hr in CTRL at 10 m above the ground (agl) in (a) and along the vertical cross section CS (cf. Figure 1a) in (b). The ucomponent of the wind is depicted as contour colors, arrows indicate the u- and v-component in (a) and u- and w-component in (b). The dashed contour line in (a) indicates a friction velocity of 0.23 m s⁻¹, black contour lines in (b) depict potential temperature with an interval of 2 K.

After the spin-up, a steady-state wind field establishes with an accelerated near-surface flow in the lee of the summit, and gravity waves propagating both vertically and horizontally (Figure 2). The main accelerated flow separates from the surface about 2 km downstream of the summit. A slight blocking upstream of the hill and a dominant zone of reversed flow downstream of the hill are present, as well as local wind speed maxima downstream of the northern and southern flanks of the hill. With the acceleration of the flow in the summit region the shear stress on the surface exceeds the threshold value of 0.23 m s⁻¹, which is u_{th} for $\rho_{snow} = 100$ kg m⁻³ based on Equation 11. Thus, we can expect snow to be eroded in this region.

The flow features described here are in line with what we can expect based on the results of the aforementioned studies. The simulations are initialized with vertically constant profiles of wind and static stability, which combine to a non-dimensional mountain height of $\hat{h} = 1$. The horizontal wave number of the terrain k ($k \approx \frac{1}{a}$, Lin, 2007) is smaller than the Scorer parameter $l \approx \frac{N_d}{U}$, thus allowing for vertically propagating gravity waves. At $\hat{h} = 1$ the flow is at the transition of establishing non-linear features such as steepening of waves, down-slope wind storms, flow blocking and reversed flows (Lin, 2007). With the accelerated flow over the summit, and the strong flow at the flanks and downstream of the flanks, the wind field shows features of both a flow-over and a flow-around regime, marking the transition between the two regimes around $\hat{h} = 1$ (Jackson et al., 2013).

In summary, the inclusion of the snow drift module to our simulation reproduces the main expected flow features and provides an environment that allows for drifting snow.

4.2. Snow Drift Patterns

The atmospheric fields govern the spatio-temporal patterns of snow redistribution. Here we will revisit this dependency, as it provides fundamental insights into the inherent behavior of the snow drift scheme. The sensitivity of the snow drift parameters will then be explained in more detail in Section 4.4. We start with the discussion on the erosion flux and then further address both the deposition and sublimation fluxes, followed by a short comparison to reported real-world drifting snow concentrations.

Based on Equation 10 we expect the erosion flux to be proportional to the excess of surface shear stress. This expectation is confirmed in Figure 3a, where erosion only takes place in regions where the friction velocity exceeds 0.23 m s^{-1} (dashed contour line, 0.23 m s^{-1} is the threshold friction velocity for a snow density of 100 kg m⁻³ based on Equation 11). Apart from that, the erosion follows the structure of the near-surface wind speed as seen in Figure 2 with the maximum erosion (about 2 kg m⁻³ in 3 hr) in the region of highest wind speeds directly in the lee of the summit. In the same way, the snow concentration in the saltation layer is dominated by the surface shear stress (Figure 3b) with maximum concentrations of about $5 \cdot 10^{-3}$ kg m⁻³ in the area of highest wind speeds. In contrast to that, more processes play a role when considering the snow concentration in suspension. Snow particles are eroded from the surface and, thus, have their maximum concentration in the lowest layer. However, turbulent motions also mix the particles within a layer of approximately 100 m (Figure 3d). Apart from that, snow particles in suspension are affected by sublimation, subsidence and advection with the mean flow (see





Figure 3. Snow mass change due to snow drift in CTRL accumulated from model time 3-6 hr as color contours (a). Arrows in (a) depict the horizontal wind field at 10 m agl as in Figure 2, the dashed contour line indicates a mean friction velocity of 0.23 m s⁻¹. Color contours depict mean snow concentration in the saltation layer (b), in 10 m agl (c) and along the vertical cross section CS (d, cf. Figure 1a).

Equation 1). Once steady-state is reached in the simulation, these processes balance each other, resulting in the suspended snow concentration as shown in Figure 3c. Here, the snow concentration does not have its maximum in the region of strongest erosion, but gradually increases in flow direction reaching its maximum of about $1.5 \cdot 10^{-3}$ kg m⁻³ at the end of the erosion zone. While the snow particles are confined to a shallow layer in the erosion zone where the vertical transport is dominated by turbulence, they are advected further away from the surface in the region of flow separation (Figure 3d).

Net deposition can only occur when the friction velocity drops below the friction velocity threshold, provided that snow particles are present in the atmosphere and the downward flux due to gravity exceeds the vertical turbulent mixing. In our case, these conditions are met directly downstream of the zone of flow separation, resulting in a narrow strip where snow is deposited (positive values in Figure 3a). Comparing the total amount of accumulated erosion to the deposition, only about 3% of the eroded mass is deposited, while the rest is removed by sublimation and suspended by the prevailing flow.

Depending on the atmospheric conditions, snow particles lose mass due to sublimation. In our implementation, several factors determine the amount of sublimation: the available snow concentration, the particle size, the particle ventilation, as well as the air temperature and relative humidity. The influence of the particle concentration and relative humidity can be seen in the spatial distribution of accumulated sublimation in Figure 4. The highest sublimation amounts occur in the region of maximum particle concentration in the lee of the summit. Presumably also the warming and thus drying of the air due to subsidence and entrainment of air from above will





Figure 4. Accumulated drifting snow sublimation in CTRL: vertically integrated (a) and local along the vertical cross section CS (b, cf. Figure 1a). Arrows in (a) depict the horizontal wind field at 10 m agl as in Figure 2. Thin (thick) contour lines in (b) indicate mean relative humidity of 95 (99)%.

enhance the sublimation here. On the other hand, the high relative humidity in the summit region inhibits sublimation there compared to regions at the northern and southern flanks of the hill where despite the lower snow concentration more sublimation is taking place (Figures 3c and 4a). In the experiments for parameter sensitivity in Section 4.4 we will explore further influences of particle size and ventilation speed.

While keeping in mind the idealized nature of the simulation presented above, a brief discussion of our results in the context of previous field measurements seems appropriate. With a simulated snow concentration in the saltation layer on the order of $5 \cdot 10^{-3}$ kg m⁻³ (Figure 3b) our results roughly agree with concentrations reported by J. Pomeroy and Male (1992) or Doorschot et al. (2004) on the order of $10^{-3}-10^{-2}$ kg m⁻³ with comparable near-surface wind velocities. Our maximum snow concentration in the lowest layer of suspension (about $1.5 \cdot 10^{-3}$ kg m⁻³, Figure 3c) are rather high compared to measurements of J. Pomeroy and Male (1992) or Vionnet et al. (2014), who report concentrations on the order of $10^{-5}-10^{-3}$ kg m⁻³ at heights on the order of 1 m above the ground. Apart from this comparison, however, one has to keep in mind that the lowest mass level of our WRF domain is located at about 5 m above the ground and is representing a layer of roughly 10 m depth. The vertical structure with strong gradients in drifting snow concentration within the lower couple of meters above the ground can therefore not be accurately depicted and horizontal snow transport here is rather overestimated.

4.3. Local and Non-Local Effect of Drifting Snow Sublimation

In order to get deeper insights into the model behavior, sensitivity experiments were conducted, where each time one model parameter or setting was alternated, while the other ones were held constant. These experiments are summarized in Table 2. In this section we concentrate on the specific role of sublimation and its feedback onto the atmosphere (NO_SUBL: switched off entire sublimation process and NO_SFB: switched off sublimation feedback). In the next section, we will present the remaining experiments. As seen before in CTRL only 3% of the eroded snow mass is deposited on the ground while the rest is suspended or sublimated. Locally, the sublimation from drifting snow particles lowers the snow concentration, increases the specific humidity and decreases the temperature of the ambient air (Figures 5a-5c). That leads to a damping of the sublimation process as this is strongly controlled by relative humidity (see Section 4.2). These effects become evident when comparing the results of the CTRL simulation to the ones of NO_SUBL and NO_SFB. Switching off sublimation leads to a drastic increase in the downstream transport of snow, as well as in the area and the amount of snow deposition (Figure 6a, Table 2: about 11 times more deposition). Apart from that, snow also gets transported out of the model domain in NO_SUBL (not shown). Compared to CTRL a slight decrease of about 4.7% in snow erosion can be seen in NO_SUBL (Table 2). This decrease in erosion can be explained by two factors: on the one hand, removing snow by sublimation increases the gradient in Equation 16 allowing for a stronger erosion. On the other hand, as will be seen later, the sublimational cooling leads to higher wind speeds close to the ground, which as well results



Table 2	2
---------	---

Summary of Sensitivity Experiments

Experiment	Model parameter	Value	Value in CTRL	Δ erosion (%)	Δ deposition (%)		
NO_SUBL		0	2	-4.7	+1025.5		
	subl_opt						
NO_SFB		1	2	-4.7	-99.8		
	subl_opt						
RHO_200		200 kg m^{-3}	100 kg m^{-3}	-34.9	-35.0		
	ρ_{snow}						
ESALT_2		10^{-3}	$5 \cdot 10^{-4}$	+106.8	+713.5		
	esalt						
ESALT_05		$2.5 \cdot 10^{-4}$	$5 \cdot 10^{-4}$	-51.7	-92.0		
	esalt						
V_01		0.1 m s^{-1}	-	-0.4	+145.2		
	pavel						
V_02		0.2 m s^{-1}	-	+0.5	+401.4		
	pavel						
R_1E4		$10^{-4} {\rm m}$	-	-1.4	+743.0		
	rgro						
R_5E5		$5 \cdot 10^{-5}$ m	-	+0.8	+106.7		
	rgro						

in a stronger erosion in the CTRL simulation. When switching off the sublimation feedback, the aforementioned damping effect no longer exists. Thus, more snow gets sublimated, leading to smaller snow concentrations (Figure 5a) and almost no deposition of snow in NO_SFB (Figure 6b).

Although the sublimation process only takes place within a shallow layer in the summit region of the hill where snow particles are present, the effects of it can be observed over a much larger region. The direct cooling and moistening effect is visible in the entire downstream region where the air affected by sublimation is transported to (Figures 7a and 7b, below 500 m). Apart from that, the gravity wave field is affected, leading to an amplification of the gravity waves, intensified up- and downdrafts as well as increased horizontal winds (Figures 7a, 7c, and 7d).

Despite the very different approaches to model the atmospheric fields, our observed local effects agree well with the results of Groot Zwaaftink et al. (2011). In both cases sublimation reduces the deposited snow mass, while the cooling and moistening dampens the sublimation process. Their approach, however, did not allow to represent the non-local response of the wind field to the local modifications. In conclusion, our results underline the findings of



Figure 5. Mean vertical profiles of suspended snow concentration (a), potential temperature (b), specific humidity (c) and u-component of the wind (d) at point P (cf. Figure 1a) for simulations CTRL, NO_SUBL and NO_SFB, averaged between model time 3–6 hr. Note that the profiles for NO_SUBL and NO_SFB overlap in (b–d).





Figure 6. Snow mass change due to snow drift in NO_SUBL (a) and NO_SFB (b) accumulated from model time 3-6 hr.

Groot Zwaaftink et al. (2011), that the calculation of drifting snow and the sublimation feedback should be directly included in the atmospheric model in order to represent these complex interactions.

4.4. Parameter Sensitivity

We start the analysis of our sensitivity experiments with a doubling of the snow density (RHO_200). For erosion to take place, the surface shear stress has to overcome the inertia and cohesive bonds of the snow particles. Thus, a higher snow density requires a higher threshold friction velocity (see Equation 11) and results in a lower erosion flux (see Equation 10). When increasing the snow density to 200 kg m⁻³ this results in an overall reduction of about 35% in both the eroded and deposited snow mass compared to CTRL (Figure 8a, Table 2). Apart from the



Figure 7. Vertical cross sections (CS, cf. Figure 1a) of the difference in mean potential temperature (a), specific humidity (b), u-component (c), and w-component of the wind (d) between CRTL and NO_SUBL, averaged between model time 3 and 6 hr (color contours). Mean potential temperature in CTRL is depicted as dashed black contour lines with 2 K interval, mean snow concentration of $5 \cdot 10^{-3}$ kg m⁻³ as a thick black contour line.





Figure 8. Accumulated snow mass change due to snow drift for sensitivity experiments RHO_200 (a), ESALT_2 (b), V_02 (c), V_01 (d), R_1E4 (e), and R_5E5 (f).

total amounts, also the erosion area is reduced to the regions of strongest winds, where friction velocity exceeds the now required threshold.

The efficiency of the erosion process is represented by the saltation parameter e_{salt} (Equation 10), which is set by default to $5 \cdot 10^{-4}$ following Sauter et al. (2013). Doubling the efficiency to $e_{salt} = 10^{-3}$ (ESALT_2) results in an approximate doubling of the accumulated eroded snow mass (Table 2). However, a non-linear response is apparent in the snow deposition, which is higher by a factor of about eight compared to CTRL. This non-linearity is a result of the self-limiting damping effect in the sublimation. As seen in Section 4.3 sublimation increases the relative humidity of the ambient air, which in return dampens subsequent sublimation. Consequently, in an almost saturated environment a further increase in snow concentration only results in a small sublimation response but a stronger downstream transport and an increased deposition of snow (Figure 8b). A further sensitivity run with e_{salt} reduced by a factor of 0.5 (ESALT_05) confirms this behavior: the erosion is reduced by about 52% while the deposition is reduced by 92% due to the stronger sublimation.

The experiments with fixed ground-level particle radii (R_1E4 and R_5E5) and fall speeds (V_01 and V_02) show the complex interplay of different processes in the model. In general, the particle size and fall speed can affect the model in several different ways. First, the fall speed is a function of particle size which on the one hand influences the downward flux of snow in suspension and thus the vertical distribution of snow. On the other hand, the ventilation speed of the snow particles that partly controls the sublimation is assumed to be equal to the particle fall speed (Sauter et al., 2013). Furthermore, the sublimation-loss rate ψ_s is proportional to r^{-2} (Sauter et al., 2013) indicating that smaller particles are more prone to sublimation. The complex interplay of these factors is apparent in the non-linear response in the experiments with fixed fall speeds and ground-level particle radii. The faster downward flux outweighs the effect of increased ventilation speed when comparing V_02 and V_01.



10.1029/2023MS004007



Figure 9. Real-case model output from the lowest model level of accumulated total snowdrift since model initialization (colors) and horizontal 10 m wind vectors (arrows) from 8 February 2021 at eight different times. The black contours show the model topography, while the blue lines show the glacier outlines as they are represented in the model domain, data from Voordendag et al. (2023).

Larger particles are falling out more quickly and are less prone to sublimation leading to higher deposition (see R_1E4, R_5E5).

The experiments also show a weakness of the default setup. Here, the ground-level particle radius is calculated as a function of friction velocity (Equation 6) and is then given to Equation 5 to calculate the height-dependent particle radius. In a situation, where the flow detaches from the ground like in our case, this results in a drastic decrease in particle size and fallout velocity for particles that are actually decoupled from the ground. As a consequence, particle deposition is drastically decreased and sublimation increased. This problem can not occur if the ground-level particle radius is fixed and is less pronounced if the fallout velocity is fixed. We therefore recommend future users to think about fixing the ground-level particle radius if the expected flow situation is very complex, for example, in mountainous environments. Possible values might be taken from the size spectra of for example, Aksamit and Pomeroy (2016), Gordon et al. (2010), Naaim-Bouvet et al. (2011), Schmidt (1981, 1986) or Vionnet et al. (2014) with average sizes on the order of 10^{-4} m. In other, less complex flow situations for example, over large ice bodies or in flat terrain the added benefit of the friction-velocity dependent particle size might still outweigh the other effects.

4.5. Outlook: Toward Real-Case Applications

Asides from idealized numerical studies, the newly implemented snow drift module can be also used for real case simulations. This application bears large potential for snow redistribution process studies in complex topography, where observations are usually sparse. An example for an area of interest embedded in complex topography with an interest in snow redistribution process is the Hintereisferner (HEF) glacier, located in the Ötztal Alps, Austria. In an accompanying study (Voordendag et al., 2023) we apply the snowdrift module to a case study of a drifting snow event on 8 February 2021 to assess model performance with a real-case large-eddy simulation setup ($\Delta x = 48 \text{ m}$) introduced by Goger et al. (2022). Snow was initialized at the same time as the atmospheric model with the standard settings of the land-surface model NOAH-MP (Niu et al., 2011) assuming a fresh snowpack, mostly related to constraints in computational resources. The event is characterized by fresh snow fall followed by strong winds (more than 10 m s⁻¹), which provides excellent conditions for drifting snow. While the majority of the



study results are outlined in Voordendag et al. (2023), we show an overview of the simulated patterns of winddriven snow redistribution in the area of interest (Figure 9). The results suggest that the modeled snow depth change due to snowdrift is strongly related to the wind field and the underlying topography, as expected for this high mountain environment. In our study, snow erosion is the major contributor to snowdrift and the process removes on average 3.9 kg m^{-3} of snow throughout the 24 hr simulation over all glaciated areas in the domain. The overall magnitude and patterns of snow redistribution are reproduced by the model reasonably well, while of course, fine-scale structures, that are smaller than the resolution of the model are missed. A more detailed analysis of the results and a detailed model evaluation with observations (Terrestrial Laser Scanning acquisitions, meteorological stations, etc.) can be found in Voordendag et al. (2023).

5. Conclusions and Outlook

In this paper we introduced a novel module to calculate drifting snow within the framework of the Weather Research and Forecasting (WRF) model. The basic idea was to have a computationally efficient module of intermediate complexity, which utilizes the existing capabilities of WRF as much as possible. For that purpose, the module is coupled to the snow scheme of the Land-Surface Model Noah-MP and employes WRF-internal modules for advection and turbulent diffusion in order to calculate the transport of drifting snow particles. The module is included directly into the calculations of WRF and, thus, is able to make use of maximum-quality wind fields and to allow for interactions of drifting snow with the background atmosphere, for example, through drifting snow sublimation. Test simulations within an idealized setting showed the physical consistency of the module. We were able to demonstrate that drifting snow sublimation influences both erosion and deposition patterns in a physically consistent way. Apart from that, we could show that the local cooling and moistening effect of drifting snow sublimation can have larger-scale non-local effects by alternating the gravity wave structure of the flow field. This underlines the necessity to include the computation of drifting snow and its feedback on the background flow directly within the atmospheric model. Due to the idealized nature of our main experiments we can only partly independently validate our results. Magnitudes of simulated snow concentrations roughly agree with field observations, while the simulated sublimation and the feedback onto the atmosphere can not be validate against field data. However, the first real-case simulations of Voordendag et al. (2023) using our snow drift module show an overall good agreement with observations. In future development steps we plan to include a radiation term into the calculation of drifting snow sublimation and to introduce a more sophisticated particle size distribution. Moreover, we intend to include the snow drift scheme into a coupling of WRF with the snow and ice model COSIPY (Sauter et al., 2020), which is currently under development. With its efficient design and incorporation into a widely used atmospheric model the module shows good potential to be used by a larger community in future studies on drifting snow for example, in mountain areas or over large ice bodies.

Data Availability Statement

The source code of the snow drift module and namelist files are provided under Saigger (2024).

References

- Abrahim, B. N., Cullen, N. J., Conway, J. P., & Sirguey, P. (2023). Applying a distributed mass-balance model to identify uncertainties in glaciological mass balance on Brewster Glacier, New Zealand. *Journal of Glaciology*, 69(276), 1–17. https://doi.org/10.1017/jog.2022.123 Aksamit, N. O., & Pomeroy, J. W. (2016). Near-surface snow particle dynamics from particle tracking velocimetry and turbulence measurements during alpine blowing snow storms. *The Cryosphere*, 10(6), 3043–3062. https://doi.org/10.5194/tc-10-3043-2016
- Amory, C., Kittel, C., Le Toumelin, L., Agosta, C., Delhasse, A., Favier, V., & Fettweis, X. (2021). Performance of MAR (v3.11) in simulating the drifting-snow climate and surface mass balance of Adélie Land, East Antarctica. *Geoscientific Model Development*, 14(6), 3487–3510. https:// doi.org/10.5194/gmd-14-3487-2021
- Arduini, G., Balsamo, G., Dutra, E., Day, J. J., Sandu, I., Boussetta, S., & Haiden, T. (2019). Impact of a multi-layer snow scheme on near-surface weather forecasts. *Journal of Advances in Modeling Earth Systems*, 11(12), 4687–4710. https://doi.org/10.1029/2019MS001725
- Bernhardt, M., Schulz, K., Liston, G., & Zängl, G. (2012). The influence of lateral snow redistribution processes on snow melt and sublimation in alpine regions. *Journal of Hydrology*, 424–425, 196–206. https://doi.org/10.1016/j.jhydrol.2012.01.001
- Beyers, J., Sundsbø, P., & Harms, T. (2004). Numerical simulation of three-dimensional, transient snow drifting around a cube. Journal of Wind Engineering and Industrial Aerodynamics, 92(9), 725–747. https://doi.org/10.1016/j.jweia.2004.03.011
- Bintanja, R. (2000). Snowdrift suspension and atmospheric turbulence. Part I: Theoretical background and model description. *Boundary-Layer Meteorology*, 95(3), 343–368. https://doi.org/10.1023/A:1002676804487
- Dadic, R., Mott, R., Lehning, M., & Burlando, P. (2010). Wind influence on snow depth distribution and accumulation over glaciers. *Journal of Geophysical Research*, 115(F1). https://doi.org/10.1029/2009JF001261
- Déry, S. J., & Yau, M. K. (1999). A bulk blowing snow model. Boundary-Layer Meteorology, 93(2), 237–251. https://doi.org/10.1023/A: 1002065615856

19422466, 2024, 6, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023MS004007 by ETH Zurich, Wiley Online Library on [12:06/2024]. See the Terms

and Conditions

brary.wiley.com/terms

-and-conditions) on Wiley Online Library for rules of use; OA articles

are governed by the applicable Creative Commons License

Acknowledgments

MS was funded by Elite Network of Bavaria-Bavarian State Ministy of Science and Art (Grand reference: IDP M3OCCA). CS was funded by the German Federal Ministry of Education and Research (BMBF, Grant 0F0778F). BG and AV were funded by the project "Measuring and modeling snow-cover dynamics at high resolution for improving distributed mass balance research on mountain glaciers," a joint project fully funded by the Austrian Science Foundation (FWF: project number I 3841-N32) and the Deutsche Forschungsgemeinschaft (DFG; project number SA 2339/7-1). The authors gratefully acknowledge the scientific support and HPC resources provided by the Erlangen National High Performance Computing Center (NHR@FAU) of the Friedrich-Alexander-Universität Erlangen-Nürnberg (FAU). The hardware is funded by the German Research Foundation (DFG). We acknowledge financial support by Deutsche Forschungsgemeinschaft and Friedrich-Alexander-Universität Erlangen-Nürnberg within the funding programme "Open Access Publication Funding." We want to thank the editor for handling our manuscript and Alexandra Gossart. Christophe Genthon and one anonymous reviewer for their constructive comments leading to the improvement of the manuscript. Open Access funding enabled and organized by Projekt DEAL.



- Déry, S. J., & Yau, M. K. (2001). Simulation of blowing snow in the Canadian Arctic using A double-moment model. Boundary-Layer Meteorology, 99(2), 297–316. https://doi.org/10.1023/A:1018965008049
- Doorschot, J. J., Lehning, M., & Vrouwe, A. (2004). Field measurements of snow-drift threshold and mass fluxes, and related model simulations. Boundary-Layer Meteorology, 113(3), 347–368. https://doi.org/10.1007/s10546-004-8659-z
- Durand, Y., Guyomarc'h, G., Mérindol, L., & Corripio, J. G. (2005). Improvement of a numerical snow drift model and field validation. Cold Regions Science and Technology, 43(1–2), 93–103. https://doi.org/10.1016/j.coldregions.2005.05.008
- Durran, D. R. (1990). Mountain waves and downslope winds. In W. Blumen (Ed.), Atmospheric processes over complex terrain (pp. 59–81). American Meteorological Society. https://doi.org/10.1007/978-1-935704-25-6_4
- Fairall, C. W., Bradley, E. F., Hare, J. E., Grachev, A. A., & Edson, J. B. (2003). Bulk parametrization of air-sea fluxes: Updates and verification for the COARE algorithm. *Journal of Climate*, 16(4), 571–591. https://doi.org/10.1175/1520-0442(2003)016(0571:BPOASF)2.0.CO;2
- Forthofer, J. M., Butler, B. W., & Wagenbrenner, N. S. (2014). A comparison of three approaches for simulating fine-scale surface winds in support of wildland fire management. Part I. Model formulation and comparison against measurements. *International Journal of Wildland Fire*, 23(7), 969. https://doi.org/10.1071/WF12089
- Gallée, H., Guyomarc'h, G., & Brun, E. (2001). Impact of snow drift on the Antarctic ice sheet surface mass balance: Possible sensitivity to snowsurface properties. *Boundary-Layer Meteorology*, 99(1), 1–19. https://doi.org/10.1023/A:1018776422809
- Gauer, P. (2001). Numerical modeling of blowing and drifting snow in Alpine terrain. Journal of Glaciology, 47(156), 97–110. https://doi.org/10. 3189/172756501781832476
- Gerber, F., Besic, N., Sharma, V., Mott, R., Daniels, M., Gabella, M., et al. (2018). Spatial variability in snow precipitation and accumulation in COSMO–WRF simulations and radar estimations over complex terrain. *The Cryosphere*, 12(10), 3137–3160. https://doi.org/10.5194/tc-12-3137-2018
- Gerber, F., Mott, R., & Lehning, M. (2019). The importance of near-surface winter precipitation processes in complex Alpine terrain. Journal of Hydrometeorology, 20(2), 177–196. https://doi.org/10.1175/JHM-D-18-0055.1
- Goger, B., Stiperski, I., Nicholson, L., & Sauter, T. (2022). Large-eddy simulations of the atmospheric boundary layer over an Alpine glacier: Impact of synoptic flow direction and governing processes. *Quarterly Journal of the Royal Meteorological Society*, 148(744), 1319–1343. https://doi.org/10.1002/qj.4263
- Gordon, M., Biswas, S., Taylor, P. A., Hanesiak, J., Albarran-Melzer, M., & Fargey, S. (2010). Measurements of drifting and blowing snow at Iqaluit, Nunavut, Canada during the star project. *Atmosphere-Ocean*, 48(2), 81–100. https://doi.org/10.3137/AO1105.2010
- Groot Zwaaftink, C. D., Löwe, H., Mott, R., Bavay, M., & Lehning, M. (2011). Drifting snow sublimation: A high-resolution 3-D model with temperature and moisture feedbacks. *Journal of Geophysical Research*, 116(D16), D16107. https://doi.org/10.1029/2011JD015754
- Groot Zwaaftink, C. D., Mott, R., & Lehning, M. (2013). Seasonal simulation of drifting snow sublimation in Alpine terrain. Water Resources Research, 49(3), 1581–1590. https://doi.org/10.1002/wrcr.20137
- Haddjeri, A., Baron, M., Lafaysse, M., Le Toumelin, L., Deschamp-Berger, C., Vionnet, V., et al. (2023). Exploring the sensitivity to precipitation, blowing snow, and horizontal resolution of the spatial distribution of simulated snow cover. *EGUsphere*, 2023, 1–44. https://doi.org/10. 5194/egusphere-2023-2604

Houze, R. A. (2012). Orographic effects on precipitating clouds. *Reviews of Geophysics*, 50(1), RG1001. https://doi.org/10.1029/2011RG000365 Jackson, P. J., Mayr, G., & Vosper, S. (2013). Dynamically-driven winds. In *Mountain weather research and forecasting: Recent progress and current challenges* (pp. 121–218). Springer. https://doi.org/10.1007/978-94-007-4098-3

- Jaedicke, C. (2002). Snow drift losses from an Arctic catchment on Spitsbergen: An additional process in the water balance. Cold Regions Science and Technology, 34(1), 1–10. https://doi.org/10.1016/S0165-232X(01)00041-6
- Jiang, Q. (2003). Moist dynamics and orographic precipitation. *Tellus A: Dynamic Meteorology and Oceanography*, 55(4), 301–316. https://doi.org/10.3402/tellusa.v55i4.14577
- Kirshbaum, D. J., & Durran, D. R. (2004). Factors governing cellular convection in orographic precipitation. Journal of the Atmospheric Sciences, 61(6), 682–698. https://doi.org/10.1175/1520-0469(2004)061(0682:FGCCIO)2.0.CO;2
- Lehning, M., Löwe, H., Ryser, M., & Raderschall, N. (2008). Inhomogeneous precipitation distribution and snow transport in steep terrain. Water Resources Research, 44(7). https://doi.org/10.1029/2007WR006545
- Lenaerts, J., & Van den Broeke, M. (2012). Modeling drifting snow in Antarctica with a regional climate model: 2. Results. Journal of Geophysical Research, 117(D5). https://doi.org/10.1029/2010JD015419
- Lenaerts, J., van den Broeke, M. R., Déry, S. J., König-Langlo, G., Ettema, J., & Munneke, P. K. (2010). Modelling snowdrift sublimation on an Antarctic ice shelf. *The Cryosphere*, 4(2), 179–190. https://doi.org/10.5194/tc-4-179-2010
- Le Toumelin, L., Amory, C., Favier, V., Kittel, C., Hofer, S., Fettweis, X., et al. (2021). Sensitivity of the surface energy budget to drifting snow as simulated by MAR in coastal Adelie Land, Antarctica. *The Cryosphere*, 15(8), 3595–3614. https://doi.org/10.5194/tc-15-3595-2021 Lin, Y.-L. (2007). *Mesoscale dynamics*. Cambridge University Press. https://doi.org/10.1017/CBO9780511619649
- Liston, G. E., Haehnel, R. B., Sturm, M., Hiemstra, C. A., Berezovskaya, S., & Tabler, R. D. (2007). Simulating complex snow distributions in windy environments using SnowTran-3D. *Journal of Glaciology*, 53(181), 241–256. https://doi.org/10.3189/172756507782202865
- Liston, G. E., & Sturm, M. (1998). A snow-transport model for complex terrain. *Journal of Glaciology*, 44(148), 498–516. https://doi.org/10.3189/ S0022143000002021
- MacDonald, M. K., Pomeroy, J. W., & Pietroniro, A. (2010). On the importance of sublimation to an alpine snow mass balance in the Canadian Rocky Mountains. *Hydrology and Earth System Sciences*, 14(7), 1401–1415. https://doi.org/10.5194/hess-14-1401-2010
- Marsh, C. B., Pomeroy, J. W., Spiteri, R. J., & Wheater, H. S. (2020). A finite volume blowing snow model for use with variable resolution meshes. *Water Resources Research*, 56(2). https://doi.org/10.1029/2019WR025307
- Marsh, C. B., Pomeroy, J. W., & Wheater, H. S. (2020). The Canadian hydrological model (CHM) v1.0: A multi-scale, multi-extent, variablecomplexity hydrological model—Design and overview. *Geoscientific Model Development*, 13(1), 225–247. https://doi.org/10.5194/gmd-13-225-2020
- Marsh, C. B., Vionnet, V., & Pomeroy, J. W. (2023). Windmapper: An efficient wind downscaling method for hydrological models. *Water Resources Research*, 59(3), e2022WR032683. https://doi.org/10.1029/2022WR032683
- Mölg, T., Chiang, J. C. H., Gohm, A., & Cullen, N. J. (2009). Temporal precipitation variability versus altitude on a tropical high mountain: Observations and mesoscale atmospheric modelling. *Quarterly Journal of the Royal Meteorological Society*, 135(643), 1439–1455. https://doi. org/10.1002/qj.461
- Mortezapour, M., Menounos, B., Jackson, P. L., Erler, A. R., & Pelto, B. M. (2020). The role of meteorological forcing and snow model complexity in winter glacier mass balance estimation, Columbia River basin, Canada. *Hydrological Processes*, 34(25), 5085–5103. https://doi. org/10.1002/hyp.13929



- Mott, R., & Lehning, M. (2010). Meteorological modeling of very high-resolution wind fields and snow deposition for mountains. Journal of Hydrometeorology, 11(4), 934–949. https://doi.org/10.1175/2010JHM1216.1
- Mott, R., Schirmer, M., Bavay, M., Grünewald, T., & Lehning, M. (2010). Understanding snow-transport processes shaping the mountain snowcover. *The Cryosphere*, 4(4), 545–559. https://doi.org/10.5194/tc-4-545-2010
- Mott, R., Schirmer, M., & Lehning, M. (2011). Scaling properties of wind and snow depth distribution in an Alpine catchment. Journal of Geophysical Research, 116(D6), D06106. https://doi.org/10.1029/2010JD014886
- Mott, R., Scipión, D., Schneebeli, M., Dawes, N., Berne, A., & Lehning, M. (2014). Orographic effects on snow deposition patterns in mountainous terrain. Journal of Geophysical Research: Atmospheres, 119(3), 1419–1439. https://doi.org/10.1002/2013JD019880
- Mott, R., Vionnet, V., & Grünewald, T. (2018). The seasonal snow cover dynamics: Review on wind-driven coupling processes. Frontiers in Earth Science, 6, 197. https://doi.org/10.3389/feart.2018.00197
- Musselman, K. N., Pomeroy, J. W., Essery, R. L., & Leroux, N. (2015). Impact of windflow calculations on simulations of alpine snow accumulation, redistribution and ablation. *Hydrological Processes*, 29(18), 3983–3999. https://doi.org/10.1002/hyp.10595
- Naaim, M., Naaim-Bouvet, F., & Martinez, H. (1998). Numerical simulation of drifting snow: Erosion and deposition models. Annals of Glaciology, 26, 191–196. https://doi.org/10.3189/1998AoG26-1-191-196
- Naaim-Bouvet, F., Naaim, M., Bellot, H., & Nishimura, K. (2011). Wind and drifting-snow gust factor in an Alpine context. Annals of Glaciology, 52(58), 223–230. https://doi.org/10.3189/172756411797252112
- Nakanishi, M., & Niino, H. (2006). An improved Mellor–Yamada level-3 model: Its numerical stability and application to a regional prediction of advection fog. *Boundary-Layer Meteorology*, 119(2), 397–407. https://doi.org/10.1007/s10546-005-9030-8
- Nappo, C. J. (2013). An introduction to atmospheric gravity waves. Academic press.
- Niu, G.-Y., Yang, Z.-L., Mitchell, K. E., Chen, F., Ek, M. B., Barlage, M., et al. (2011). The community Noah land surface model with multiparameterization options (Noah-MP): 1. Model description and evaluation with local-scale measurements. *Journal of Geophysical Research*, 116(D12), D12109. https://doi.org/10.1029/2010JD015139
- Pomeroy, J., & Gray, D. (1990). Saltation of snow. Water Resources Research, 26(7), 1583–1594. https://doi.org/10.1029/WR026i007p01583
- Pomeroy, J., Gray, D., & Landine, P. (1993). The prairie blowing snow model: Characteristics, validation, operation. Journal of Hydrology, 144(1-4), 165–192. https://doi.org/10.1016/0022-1694(93)90171-5
- Pomeroy, J., & Male, D. (1992). Steady-state suspension of snow. *Journal of Hydrology*, *136*(1–4), 275–301. https://doi.org/10.1016/0022-1694 (92)90015-n
- Pomeroy, J. W., & Gray, D. M. (1995). Snow accumulation, relocation and management Science Report No. 7. National Hydrology Research Institute. Environment Canada: Saskatoon.
- Pomeroy, J. W., Gray, D. M., Shook, K. R., Toth, B., Essery, R. L. H., Pietroniro, A., & Hedstrom, N. (1998). An evaluation of snow accumulation and ablation processes for land surface modelling. *Hydrological Processes*, 12(15), 2339–2367. https://doi.org/10.1002/(SICI)1099-1085 (199812)12:15(2339::AID-HYP800)3.0.CO;2-L
- Powers, J. G., Klemp, J. B., Skamarock, W. C., Davis, C. A., Dudhia, J., Gill, D. O., et al. (2017). The weather research and forecasting model: Overview, system efforts, and future directions. *Bulletin of the American Meteorological Society*, 98(8), 1717–1737. https://doi.org/10.1175/ BAMS-D-15-00308.1
- Pradhananga, D., & Pomeroy, J. W. (2022). Diagnosing changes in glacier hydrology from physical principles using a hydrological model with snow redistribution, sublimation, firnification and energy balance ablation algorithms. *Journal of Hydrology*, 608, 127545. https://doi.org/10. 1016/j.jhydrol.2022.127545
- Saigger, M. (2024). WRFsnowdrift [Software]. Zenodo. https://doi.org/10.5281/zenodo.10837359
- Sauter, T., Arndt, A., & Schneider, C. (2020). COSIPY v1.3—An open-source coupled snowpack and ice surface energy and mass balance model. Geoscientific Model Development, 13(11), 5645–5662. https://doi.org/10.5194/gmd-13-5645-2020
- Sauter, T., Möller, M., Finkelnburg, R., Grabiec, M., Scherer, D., & Schneider, C. (2013). Snowdrift modelling for the Vestfonna ice cap, northeastern Svalbard. *The Cryosphere*, 7(4), 1287–1301. https://doi.org/10.5194/tc-7-1287-2013
- Schirmer, M., Wirz, V., Clifton, A., & Lehning, M. (2011). Persistence in intra-annual snow depth distribution: 1. Measurements and topographic control. Water Resources Research, 47(9). https://doi.org/10.1029/2010WR009426
- Schmidt, R. A. (1980). Threshold wind-speeds and elastic impact in snow transport. Journal of Glaciology, 26(94), 453–467. https://doi.org/10.3189/S0022143000010972
- Schmidt, R. A. (1981). Estimates of threshold windspeed from particle sizes in blowing snow. Cold Regions Science and Technology, 4(3), 187–193. https://doi.org/10.1016/0165-232X(81)90003-3
- Schmidt, R. A. (1986). Transport rate of drifting snow and the mean wind speed profile. *Boundary-Layer Meteorology*, 34(3), 213–241. https://doi.org/10.1007/BF00122380
- Schneiderbauer, S., Tschachler, T., Fischbacher, J., Hinterberger, W., & Fischer, P. (2008). Computational fluid dynamic (CFD) simulation of snowdrift in alpine environments, including a local weather model, for operational avalanche warning. *Annals of Glaciology*, 48, 150–158. https://doi.org/10.3189/172756408784700789
- Sharma, V., Gerber, F., & Lehning, M. (2023). Introducing CRYOWRF v1.0: Multiscale atmospheric flow simulations with advanced snow cover modelling. *Geoscientific Model Development*, 16(2), 719–749. https://doi.org/10.5194/gmd-16-719-2023
- Skamarock, W. C., Klemp, J. B., Dudhia, J., Gill, D. O., Liu, Z., Berner, J., et al. (2019). A description of the advanced research WRF model version 4 (Tech. Rep.). UCAR/NCAR. https://doi.org/10.5065/1DFH-6P97
- Smith, R. B. (1979). The influence of mountains on the atmosphere. Advances in Geophysics, 21, 87–230. https://doi.org/10.1016/S0065-2687 (08)60262-9
- Strasser, U., Bernhardt, M., Weber, M., Liston, G. E., & Mauser, W. (2008). Is snow sublimation important in the alpine water balance? *The Cryosphere*, 2(1), 53–66. https://doi.org/10.5194/tc-2-53-2008
- Temme, F., Farías-Barahona, D., Seehaus, T., Jaña, R., Arigony-Neto, J., Gonzalez, I., et al. (2023). Strategies for regional modeling of surface mass balance at the Monte Sarmiento Massif, Tierra del Fuego. *The Cryosphere*, 17(6), 2343–2365. https://doi.org/10.5194/tc-17-2343-2023
- Terleth, Y., van Pelt, W. J. J., & Pettersson, R. (2023). Spatial variability in winter mass balance on Storglaciären modelled with a terrain-based approach. Journal of Glaciology, 69(276), 749–761. https://doi.org/10.1017/jog.2022.96
- Thorpe, A. D., & Mason, B. J. (1966). The evaporation of ice spheres and ice crystals. *British Journal of Applied Physics*, 17(4), 541–548. https://doi.org/10.1088/0508-3443/17/4/316
- Umek, L., Gohm, A., Haid, M., Ward, H. C., & Rotach, M. W. (2021). Large-eddy simulation of foehn–cold pool interactions in the Inn Valley during PIANO IOP 2. *Quarterly Journal of the Royal Meteorological Society*, 147(735), 944–982. https://doi.org/10.1002/qj.3954
- Vionnet, V., Marsh, C. B., Menounos, B., Gascoin, S., Wayand, N. E., Shea, J., et al. (2021). Multi-scale snowdrift-permitting modelling of mountain snowpack. *The Cryosphere*, 15(2), 743–769. https://doi.org/10.5194/tc-15-743-2021



- Vionnet, V., Martin, E., Masson, V., Guyomarc'h, G., Naaim-Bouvet, F., Prokop, A., et al. (2014). Simulation of wind-induced snow transport and sublimation in alpine terrain using a fully coupled snowpack/atmosphere model. *The Cryosphere*, 8(2), 395–415. https://doi.org/10.5194/tc-8-395-2014
- Vionnet, V., Martin, E., Masson, V., Lac, C., Naaim Bouvet, F., & Guyomarc'h, G. (2017). High-resolution large eddy simulation of snow accumulation in Alpine terrain. *Journal of Geophysical Research: Atmospheres*, 122(20), 11005–11021. https://doi.org/10.1002/ 2017JD026947
- Voordendag, A., Goger, B., Prinz, R., Sauter, T., Mölg, T., Saigger, M., & Kaser, G. (2023). Investigating wind-driven snow redistribution processes over an alpine glacier with high-resolution terrestrial laser scans and large-eddy simulations. *The Cryosphere*, 18(2), 849–868. https://doi.org/10.5194/egusphere-2023-1395
- Walter, M. T., McCool, D. K., King, L. G., Molnau, M., & Campbell, G. S. (2004). Simple snowdrift model for distributed hydrological modeling. *Journal of Hydrologic Engineering*, 9(4), 280–287. https://doi.org/10.1061/(ASCE)1084-0699(2004)9:4(280)
- Wang, C.-C., & Kirshbaum, D. J. (2015). Thermally forced convection over a mountainous tropical island. *Journal of the Atmospheric Sciences*, 72(6), 2484–2506. https://doi.org/10.1175/JAS-D-14-0325.1
- Winstral, A., & Marks, D. (2002). Simulating wind fields and snow redistribution using terrain-based parameters to model snow accumulation and melt over a semi-arid mountain catchment. *Hydrological Processes*, 16(18), 3585–3603. https://doi.org/10.1002/hyp.1238
- Winstral, A., Marks, D., & Gurney, R. (2013). Simulating wind-affected snow accumulations at catchment to basin scales. Advances in Water Resources, 55, 64–79. https://doi.org/10.1016/j.advwatres.2012.08.011
- Wyngaard, J. C. (2004). Toward numerical modeling in the "Terra Incognita". Journal of the Atmospheric Sciences, 61(14), 1816–1826. https:// doi.org/10.1175/1520-0469(2004)061(1816:TNMITT)2.0.CO;2
- Xie, Z., Hu, Z., Ma, Y., Sun, G., Gu, L., Liu, S., et al. (2019). Modeling blowing snow over the Tibetan Plateau with the community land model: Method and preliminary evaluation. *Journal of Geophysical Research: Atmospheres*, 124(16), 9332–9355. https://doi.org/10.1029/ 2019JD030684
- Zängl, G. (2007). Interaction between dynamics and cloud microphysics in orographic precipitation enhancement: A high-resolution modeling study of two north alpine heavy-precipitation events. *Monthly Weather Review*, 135(8), 2817–2840. https://doi.org/10.1175/MWR3445.1
- Zängl, G. (2008). The temperature dependence of small-scale orographic precipitation enhancement. *Quarterly Journal of the Royal Meteorological Society*, 134(634), 1167–1181. https://doi.org/10.1002/qj.267